Author’s response to reviews

Title: Complicated Gitelman syndrome and autoimmune thyroid disease: A case report with a new homozygous mutation in the SLC12A3 gene and literature review

Authors:
Haiyang Zhou (luozuojie@gxmu.edu.cn)
Xinhuan Liang (lxhcnnn@163.com)
Yingfen Qing (yingfeng@126.com)
Bihui Meng (729382773@qq.com)
Jia Zhou (ab0851@163.com)
Song Huang (huangsong71@qq.com)
Shurong Lu (625818278@qq.com)
Zhenxing Huang (huangzx11@126.com)
Haiyan Yang (eversummeryoung@yahoo.net)
Yan Ma (luoma628@163.com)
Zuojie Luo (zhouhaiyang4000@163.com)

Version: 3 Date: 19 Aug 2018

Author’s response to reviews:

Dear editor

Thank you for your kind letter of “Gitelman syndrome complicated with autoimmune thyroid disease: A case report with a new homozygous mutation in the SLC12A3 gene and literature review” on July 19, 2018. We revised the articles according to the reviewers’ advices, including three amendments which had been marked as yellow.

1. We add some detail of sequencing methods for this case. The new information include the tool we used and all gene targeted (new table3).

2. We changed mutation description, legends and legend's name as the advice.
3. We realized that the mutation had been described as a compound heterozygote, so we changed our view of the first discovery and added the corresponding literature.

The following is the reply to Dr. Katarzyna Ziemnicka.

Dear Dr. Katarzyna Ziemnicka

I am very grateful to your comments for the manuscript. According with your advice, we amended the relevant part in manuscript.

1. We add some detail of sequencing methods for this case. The new information include the tool we used and all gene targeted (new table3).

2. We changed mutation description, legends and legend's name as the advice.

3. We realized that the mutation had been described as a compound heterozygote, so we changed our view of the first discovery and added the corresponding literature.

Thank you very much for all your help and looking forward to hearing from you soon.

Best regards

Sincerely yours Dr luo